Author's response to reviews

Title: Single Nucleotide Polymorphisms in DNA Repair Genes as Risk Factors Associated to Prostate Cancer Progression

Authors:

Luis Alberto Henriquez-Hernandez (lhenriquez@dcc.ulpgc.es)
Almudena Valenciano (almuvalenciano@gmail.com)
Palmira Foro-Arnalot (pforo@parcdesalutmar.cat)
María Jesús Álvarez-Cubero (mjesusac@ugr.es)
José Manuel Cozar (Cozarjm@yahoo.es)
José Francisco Suárez-Novo (jfsuarez@bellvitgehospital.cat)
Manel Castells-Esteve (mcastells@bellvitgehospital.cat)
Pablo Fernández-Gonzalo (pfernandez@onkologikoa.org)
Belén De-Paula-Carranza (bdepaula@onkologicoa.org)
Montse Ferrer (mferrer@imim.es)
Ferrán Guedea (guedea@iconcologia.net)
Gemma Sancho-Pardo (gsancho@santpau.cat)
Jordi Craven-Bartle (jcraven@santpau.cat)
María José Ortiz-Gordillo (mariaj.ortiz.sspa@juntadeandalucia.es)
Patricia Cabrera-Roldán (patricia.cabrera.sspa@juntadeandalucia.es)
Estefanía Herrera-Ramos (faniya1@gmail.com)
Carlos Rodríguez-Gallego (jrodrgal@gobiernodecanarias.org)
Juan Ignacio Rodríguez-Melcón (nachorodriguezmelcon@hotmail.com)
Pedro C Lara (plara@dcc.ulpgc.es)

Version: 3
Date: 15 December 2014

Author's response to reviews: see over
December 15, 2014

Dear Sir,

Thank you for accepting our original paper entitled “Single Nucleotide Polymorphisms in DNA Repair Genes as Risk Factors Associated to Prostate Cancer Progression” for publication in BMC Medical Genetics.

We have included the required changes in the new version of the manuscript. Please, find enclosed the reviewed version of the manuscript and our comments further down.

Yours sincerely,

Luis Alberto Henríquez Hernández, Ph.D.
Radiation Oncology Department.
Hospital Universitario de Gran Canaria Dr. Negrín.
Las Palmas de Gran Canaria (Spain)
EDITORIAL COMMENTS

1. Please check for typographical errors. 'Supplementary' is misspelt in supplementary file 2 and 'founders' rather than 'funders' in the Acknowledgements.

Authors’ comments
We have corrected these typographical errors. The manuscript was also checked and revised in that sense.

2. Please include details of all the ethics boards that gave approval for the study in the manuscript.

Authors’ comments
We have included this information in the Material and methods section.